Author’s response to reviews

Title: Differential Experiences of Discrimination among Ethnoracially Diverse Persons Experiencing Mental Illness and Homelessness

Authors:

Suzanne Zerger (zergers@smh.ca)
Sarah Bacon (sarah.bacon@mail.utoronto.ca)
Simon Corneau (simon.corneau@gmail.com)
Anna Skosireva (Skosanna@gmail.com)
Kwame McKenzie (Kwame.McKenzie@camh.ca)
Susan Gapka (susangapka@yahoo.ca)
Pat O'Campo (O'CampoP@smh.ca)
Aseefa Sarang (aseefa@acrossboundaries.ca)
Vicky Stergiopoulos (StergiopoulosV@smh.ca)

Version: 4 Date: 27 October 2014

Author’s response to reviews: see over
October 27, 2014

Dear Dr. Monnica Williams:

We are very pleased that you have accepted our manuscript for publication in your well-respected journal. Below we have responded to your final editorial requests.

*Editorial requests:

1. Requesting trial registration where clinical trials are described (Asking authors for TRN):

Once you know your trial registration number, please submit a revised version of your manuscript with the number included in the abstract. The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number and the date of registration, e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Registered 28 September 2004. Please note that there should be no space between the letters and numbers of the trial registration number.

2. Date of Trial Registration

The last section of the abstract should be Trial Registration: listing the trial registry and the unique identifying number and the date of registration, e.g. Trial registration: Current Controlled Trials ISRCTN73824458. Registered 28 September 2004. Please note that there should be no space between the letters and numbers of the trial registration number.

To address these requests, we have added the following to the abstract in our final version of the manuscript:

Trial registration: Current Controlled Trials ISRCTN42520374. Registered 18 August 2009.

Thank you again! Please feel free to contact me with any additional comments or questions.

Sincerely,

Vicky Stergiopoulos, corresponding author
Phone: 416.864.6060
E-mail: StergiopoulosV@smh.ca